 Vc/29-Mar-05
Week 13  starting 28  March 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  28 March 05 @ 3/28/2005 6:43:58 PM   
	1
	A1
	MBBL1196
	PT 13.2 Resist. Meas. 
	299.65

	3
	A2
	MBBL3130
	PT 6.3 Training 3
	3.95 HV

	7
	B1
	SSS125
	PT 12 Warm Up
	202.58

	12
	B2
	MBBL2157
	PT 1 IAP @ cold
	1.90

	10
	C1
	MBAL1200
	ICS 1 WP04 HV Test Warm
	294.1

	8
	C2
	MBAL2125
	PT 6.1 Training 1
	1.90

	6
	D1
	SSS028
	PT 14 Magnet MAPS
	294.91

	5
	D2
	SSS025
	PT 6.7 Training 7
	1.91

	11
	E1
	MBBR3294
	Finished Ready to go Final scan
	295.89

	4
	E2
	MBBR3312
	Finished Ready to go Final scan
	295.8

	2
	F1
	MBAR3299
	PT 13.2 Resist. Meas. 
	300.82

	9
	F2
	MBBL2162
	Prep 5 Pumping + Cool Down
	5.49 HV


Temperature data are taken from TBSM18 site
All in RED are Finished Magnets , some since  Many Hours ( even yesterday, Sunday) & there are no new magnets coming in ( ICS, MRB, Rocla etc problems over Easter: Ask Vittorio !) 
Coordinator this week: Stephane; MTM support on call :   16-3602 .

15 Magnets Tested  
11  dipoles( 1 with MM)+ 3  SSS(0  with MM) +  1  SSSS (with MM)
    Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :
General Points 
IAP  problems     R init 20.7    etc etc   …
AQA & Alarms & New Version Distribution    -- when ??
SSSS  Reporting for Strip/standby
  AOB

	


Tests 

	Magnets under Test from 20-03-2005 to 26-03-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL1184 
	[image: image1.png]



	


	14-03-2005 
	32 h.
	43 h.
	42 h.
	15 h.
	10 h.
	20-03-2005 
	142 h.
	
	

	2
	TBA2
	MBBR3275 
	[image: image3.png]



	


	18-03-2005 
	56 h.
	30 h.
	50 h.
	16 h.
	12 h.
	24-03-2005 
	164 h.
	
	

	3
	TBB1
	SSS602 
	[image: image5.png]



	


	14-03-2005 
	77 h.
	15 h.
	119 h.
	8 h.
	26 h.
	25-03-2005 
	245 h.
	
	

	4
	TBB2
	MBAL1202 
	[image: image7.png]



	


	20-03-2005 
	22 h.
	28 h.
	32 h.
	19 h.
	24 h.
	25-03-2005 
	125 h.
	
	

	5
	TBB2
	MBAR3251 
	[image: image9.png]



	


	14-03-2005 
	30 h.
	27 h.
	38 h.
	28 h.
	16 h.
	20-03-2005 
	139 h.
	
	

	6
	TBC1
	MBAL1209 
	[image: image11.png]



	


	17-03-2005 
	29 h.
	30 h.
	17 h.
	14 h.
	10 h.
	21-03-2005 
	100 h.
	
	

	7
	TBC2
	MBAL1204 
	[image: image13.png]



	


	17-03-2005 
	37 h.
	22 h.
	14 h.
	14 h.
	9 h.
	21-03-2005 
	96 h.
	
	

	8
	TBC2
	MBBL1165 
	[image: image15.png]



	


	21-03-2005 
	26 h.
	25 h.
	16 h.
	18 h.
	11 h.
	25-03-2005 
	96 h.
	
	

	9
	TBD1
	SSS103 
	[image: image17.png]



	


	18-03-2005 
	37 h.
	15 h.
	34 h.
	9 h.
	17 h.
	22-03-2005 
	112 h.
	
	

	10
	TBD1
	SSS103 
	[image: image19.png]



	


	22-03-2005 
	2 h.
	-
	-
	-
	-
	23-03-2005 
	14 h.
	
	

	11
	TBD2
	SSS019 
	[image: image21.png]



	


	20-03-2005 
	41 h.
	18 h.
	19 h.
	9 h.
	13 h.
	24-03-2005 
	100 h.
	
	

	12
	TBD2
	SSS128 
	[image: image23.png]



	


	15-03-2005 
	30 h.
	21 h.
	35 h.
	9 h.
	25 h.
	20-03-2005 
	120 h.
	
	

	13
	TBE2
	MBBR3293 
	[image: image25.png]



	


	17-03-2005 
	31 h.
	26 h.
	37 h.
	13 h.
	10 h.
	22-03-2005 
	117 h.
	
	

	14
	TBF1
	MBAL1203 
	[image: image27.png]



	


	19-03-2005 
	31 h.
	25 h.
	29 h.
	14 h.
	10 h.
	24-03-2005 
	109 h.
	
	

	15
	TBF2
	MBAL2101 
	[image: image29.png]



	


	20-03-2005 
	19 h.
	30 h.
	42 h.
	13 h.
	16 h.
	25-03-2005 
	120 h.
	
	

	16
	TBF2
	MBAR3303 
	[image: image31.png]



	


	11-03-2005 
	22 h.
	78 h.
	93 h.
	13 h.
	7 h.
	20-03-2005 
	213 h.
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL1184
	12844.1 A.
(D2-U-S01)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBR3275
	9193 A.
(D1-L-S01)
	11685 A.
(D2-L-S01)
	12322 A.
(D2-L-S01)
	12738 A.
(D2-L-S01)
	12763 A.
(D2 - U)
	12851 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-

	3
	SSS602
	5430 A.
(No Data)
	5850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL1202
	12133.5 A.
(D1-L)
	12284 A.
(D2_U)
	[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAR3251
	10417 A.
(D2-L)
	11356 A.
(D2-U)
	11980 A.
(D1-U)
	12678 A.
(D2-L)
	12618 A.
(D1-L)
	12604 A.
(D1-L)
	12750 A.
(D1_L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-

	6
	MBAL1209
	12129 A.
(D2_L)
	12513 A.
(D2_U)
	[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAL1204
	12196 A.
(D1_L)
	12386 A.
(D1-L)
	[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBL1165
	11750 A.
(D1_U)
	12393 A.
(D2-L)
	[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS103
	12000 A.
(QE-34)
	11566.3 A.
(QE-34)
	12056.3 A.
(QE_12)
	12569 A.
(QE-12)
	12686 A.
(QI_12)
	12771.2 A.
(QI-34)
	12812 A.
(QE - 12)
	12638 A.
(QE - 12)
	12850 A.
[image: image41.jpg]



	-
	-
	-
	-

	10
	SSS103
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS019
	12588 A.
(QE-34)
	12462 A.
(QE_12)
	[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	SSS128
	11852 A.
(QE - 34)
	11796 A.
(QE_34)
	12349 A.
(QI_34)
	12551 A.
(QI-12)
	12681 A.
(QE-12)
	12803 A.
(QI - 12)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-

	13
	MBBR3293
	11568.8 A.
(D2-U)
	11770.9 A.
(D1-U)
	11910 A.
(D2_L)
	12109 A.
(D2-L)
	12249 A.
(D2-U)
	12142 A.
(D2-L)
	12191.6 A.
(D2-L)
	12448.9 A.
(D1_U)
	12550 A.
(D2-U)
	[image: image45.png]



	-
	-
	-

	14
	MBAL1203
	12221.8 A.
(D1_U)
	12413 A.
(D2-L)
	[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBAL2101
	10372.1 A.
(D2_L)
	12186.6 A.
(D1_L)
	[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBAR3303
	11417 A.
(D1-L)
	11678 A.
(D1 - U)
	12130 A.
(D2-U)
	12096 A.
(D2 - U)
	12331 A.
(D2-U-S01)
	12420 A.
(D2-L)
	12287.8 A.
(D2_L)
	12610 A.
(D2_U)
	12646 A.
(D2_U)
	[image: image48.png]



	-
	-
	-


 
Week 13 march 20 to 26 March 05
[3275 waiting for 1184 or was it special MM or MM problems?
3303 long cooldown problem  or cohabitation ?  ]

[image: image49.png]Magnet Summary
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	From 20-03-2005 to 26-03-2005 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBAL2101 
	False 
	20-03-2005 
	19 h.
	30 h.
	42 h.
	13 h.
	16 h.
	25-03-2005 
	120 h.
	2 

	2
	TBF2
	MBAR3303 
	False 
	11-03-2005 
	22 h.
	78 h.
	93 h.
	13 h.
	7 h.
	20-03-2005 
	213 h.
	9 

	3
	TBF1
	MBAL1203 
	False 
	19-03-2005 
	31 h.
	25 h.
	29 h.
	14 h.
	10 h.
	24-03-2005 
	109 h.
	2 

	4
	TBE2
	MBBR3293 
	False 
	17-03-2005 
	31 h.
	26 h.
	37 h.
	13 h.
	10 h.
	22-03-2005 
	117 h.
	9 

	5
	TBD2
	SSS019 
	False 
	20-03-2005 
	41 h.
	18 h.
	19 h.
	9 h.
	13 h.
	24-03-2005 
	100 h.
	2 

	6
	TBD2
	SSS128 
	False 
	15-03-2005 
	30 h.
	21 h.
	35 h.
	9 h.
	25 h.
	20-03-2005 
	120 h.
	6 

	7
	TBD1
	SSS103 
	False 
	18-03-2005 
	37 h.
	15 h.
	34 h.
	9 h.
	17 h.
	22-03-2005 
	112 h.
	8 

	8
	TBD1
	SSS103 
	False 
	22-03-2005 
	2 h.
	137 h.
	-136 h.
	4 h.
	7 h.
	23-03-2005 
	14 h.
	0 

	9
	TBC2
	MBAL1204 
	False 
	17-03-2005 
	37 h.
	22 h.
	14 h.
	14 h.
	9 h.
	21-03-2005 
	96 h.
	2 

	10
	TBC2
	MBBL1165 
	False 
	21-03-2005 
	26 h.
	25 h.
	16 h.
	18 h.
	11 h.
	25-03-2005 
	96 h.
	2 

	11
	TBC1
	MBAL1209 
	False 
	17-03-2005 
	29 h.
	30 h.
	17 h.
	14 h.
	10 h.
	21-03-2005 
	100 h.
	2 

	12
	TBB2
	MBAL1202 
	False 
	20-03-2005 
	22 h.
	28 h.
	32 h.
	19 h.
	24 h.
	25-03-2005 
	125 h.
	2 

	13
	TBB2
	MBAR3251 
	False 
	14-03-2005 
	30 h.
	27 h.
	38 h.
	28 h.
	16 h.
	20-03-2005 
	139 h.
	7 

	14
	TBB1
	SSS602 
	True 
	14-03-2005 
	77 h.
	15 h.
	119 h.
	8 h.
	26 h.
	25-03-2005 
	245 h.
	1 

	15
	TBA2
	MBBR3275 
	True 
	18-03-2005 
	56 h.
	30 h.
	50 h.
	16 h.
	12 h.
	24-03-2005 
	164 h.
	5 

	16
	TBA1
	MBBL1184 
	False 
	14-03-2005 
	32 h.
	43 h.
	42 h.
	15 h.
	10 h.
	20-03-2005 
	142 h.
	1 


 

 

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _22_A


	TBA2
	Over-ranges in 3275 magnet during MM1

	MTM
	
	Connector not screwed properly.

	Tuesday _22_A


	TBA2
	QH discharge. Temperature of Lead B is not correct.
	ACR
	
	

	Tuesday _22_A


	TBB2
	HV test: automatic loading of the magnet name not possible.

 Error : PLC-SE does not allow to start the test.
	MTM
	HV
	Richard loaded the name manually. 
Investigation by Richard.

	Tuesday _22_N

	TBE
	Connection between PLC and HV rack MY93 not possible.
	ACR
	
	

	Wednesday _23_M


	TBF2
	Cryo OK lost.
	ACE
	
	Decrease of He liquid in the CFB

	Wednesday _23_M


	TBD2
	Corrector circuit cannot to be powered. Trigger generated every time.
	ICS
	
	Connection not correct. TC decided to end up the cold test.

	Wednesday _23_M


	TBD1
	P11 connector has been broken during magnet movement.


	MTM
	
	

	Wednesday _23_A


	TBA2
	SPA occurred on the training number 6.
	ACR?
	
	Cryo problem?

	Wednesday _23_A


	TBA1
	MITTS alert on 3 kA diode test.
	MTM
	AQA
	Pierre contacted. MIITS limit increased with the new AQA version 27 instead of 23 so the alarm became a warning.



	Thursday_24_M


	TBC2
	Training 1: Test could not be started 2 times
	MTM
	TEMA
	TEMA switch off and re-started. Then OK.


	Thursday_24_A


	TBC1
	SPA at 1 kA: unable to launch it due to interlocks “leadTempA_X1” and “LeadTempB_x1”which are not re-settable.
	ACR/AB-CO
	
	Maryline called. 

CTH is sending the signal which creates the interlocks. C.Dehavay has to decrease the thresholds of these signals.

Software will be modified by A.Tovar Tuesday.


	Thursday_24_A

Friday:25_M
	TBB2

TBE
	IAP at warm after test not launched. DAQ could not be armed. 

Same problem
	AB/CO
	Hf,lf
	Local sun server switched off/on and TEMA re-launched.


	Friday_25_M


	TBD2
	Rinit to low for QH1&4 (20 ohm)
	MTM
	hfoff
	Maryline informed. She advised to use the hfoff application of the sun MTA 24: values OK now.

	Friday_25_A


	TBD2
	SSS25 : HV test at warm: YT311 vs ground has a wrong value of capacitance.
	MTM
	
	Maryline informed. Investigation through the IFS box. 

	Friday_25_A


	TBD1
	Cryo OK lost during the powering of the EF circuit.
	ACR
	
	Cryo lost. TC advised to re-launch the test.

	Friday_25_N


	TBB1
	SSS125: QH could not be charged.
	ICS
	
	Maryline informed.Vtaps in the internal lead clamp not connected. ICS did the connection again.

	Friday_25_N


	TBA1
	1196 magnet : lf QH test : Iearth and QH signals not found.
	AB/CO
	
	Maryline advised to reset the crates before proceeding with the next quench.

	Saturday_26_N


	TBA1
	Training quench number 5: 

-lf file not present in the control room sun.

- QH timing error with AQA found in hf file
	AB/CO
	
	Analysis performed on lf file in local sun: signals OK.

	Sunday_27_M


	TBA1
	DAQ could not be loaded correctly.
	AB/CO
	
	Reset the crates but hf DAQ cannot be armed.

Reset again but then MXI communication error. Maryline changed the first card of the hf crate.

	Sunday_27_M


	TBD2
	SSS025:Trigger which prevents to power the EF correctors
	MTM
	
	Training of the MQ proceeded. Maryline is informed. She shorted the signals in the security matrix.

	Sunday_27_N


	TBB1
	HF quench at 1.5 kA on SSS125: QH3 and QH4 did not discharge. Test repeated: same result.


	MTM
	
	Maryline suggested to reset all the DAQ system (crates, lf, hf applications,): same result. Provoked quench at 2000A. Hf trigger is received and all Heaters discharged. 

AQA alarm tFire<-2000

	Sunday_27_N


	TBB1
	Powering EF circuit not possible: 
	MTM
	
	Maryline changed the inductance value~3 mH

	Monday_28_M


	TBD2
	Slow PA during training ramp 
	?
	
	


 










